Author's response to reviews

Title: Distraction osteogenesis in a severe mandibular deficiency

Authors:

- Kerim Ortakoglu (kortakoglu@yahoo.com)
- Seniz Karacay (senkaracay@yahoo.com)
- Metin Sencimen (metinsencimen@yahoo.com)
- Erol Akin (erol1akin@hotmail.com)
- Aykut H Ozyigit (aykutozyigit@yahoo.com)
- Osman Bengi (osmanbengi@hotmail.com)

Version: 8 Date: 28 December 2006

Author's response to reviews:

1. The unnecessary capitals have all been removed from the title

2. Dr Ozyigit's name has been written in the same order in both the submission system and the manuscript author list

3. Sentence case has been used for the author surnames

4. Keywords section has been removed from the manuscript

5. Introduction has been changed as background

6. Reference format has been changed as directed

7. Authors contributions has been corrected as Authors' contributions.

8. The patient's eyes has been obscured